Author’s response to reviews

Title: Advance directives as a tool to respect patients' values and preferences. Discussion on the case of Alzheimer's disease.

Authors:

Corinna Porteri (cporteri@fatebenefratelli.eu; portericorinna@gmail.com)

Version: 2 Date: 06 Feb 2018

Author’s response to reviews:

Dear Editor,

I am submitting the revised version of my manuscript entitled “Advance directives as a tool to respect patients’ values and preferences. Discussion on the case of Alzheimer’s disease”.

As required, I structured the abstract with a background, main body and conclusion; and the text with a background, main text and conclusions.

Moreover, as suggested by the first reviewer, I changed “we” in “I” throughout the abstract and the text.

I hope the work will now be suitable for publication.

Best regards,

Corinna Porteri

Brescia, 6.2.2018